# Linkage Rules for Plant–Pollinator Networks: Trait Complementarity or Exploitation Barriers?

Luis Santamaría<sup>1\*</sup>, Miguel A. Rodríguez-Gironés<sup>2</sup>

1 Mediterranean Institute for Advanced Studies, University of the Balearic Islands/Spanish Council for Scientific Research, Esporles, Mallorca, Spain, 2 Estación Experimental de Zonas Áridas, Spanish Council for Scientific Research, Almería, Spain

Recent attempts to examine the biological processes responsible for the general characteristics of mutualistic networks focus on two types of explanations: nonmatching biological attributes of species that prevent the occurrence of certain interactions ("forbidden links"), arising from trait complementarity in mutualist networks (as compared to barriers to exploitation in antagonistic ones), and random interactions among individuals that are proportional to their abundances in the observed community ("neutrality hypothesis"). We explored the consequences that simple linkage rules based on the first two hypotheses (complementarity of traits versus barriers to exploitation) had on the topology of plant–pollination networks. Independent of the linkage rules used, the inclusion of a small set of traits (two to four) sufficed to account for the complex topological patterns observed in real-world networks. Optimal performance was achieved by a "mixed model" that combined rules that link plants and pollinators whose trait ranges overlap ("complementarity models"). Deterrence of floral parasites (barrier model) is therefore at least as important as increasing pollination efficiency (complementarity model) in the evolutionary shaping of plant–pollinator networks.

Citation: Santamaría L, Rodríguez-Gironés MA (2007) Linkage rules for plant-pollinator networks: Trait complementarity or exploitation barriers? PLoS Biol 5(2): e31. doi:10. 1371/journal.pbio.0050031

# Introduction

Generalisation is a widespread feature of plant-pollinator interactions [1], and there is a growing interest in the study of complete communities of interacting plants and flowervisiting insects [2,3]. The topology of plant-pollinator qualitative networks (networks that simply consider whether species pairs interact or not) follows certain regular patterns [2-6]. The number of interactions (L) increases with network size (S, the sum of the number of plant species, F, and pollinator species, P), following a power-law relationship [4], while the percentage connectivity  $[C = 100 \times L/(F \times P)]$ , referred to as connectivity, or C, hereafter] decreases with network size [3]. Furthermore, most plant-pollinator networks are highly nested, i.e., pollinators that visit a plant species are likely to visit more-generalist plant species as well [3,4]. The nestedness (N) of a network increases with its C and size, and it is significantly higher than expected by chance for all moderately large pollination networks (i.e., with more than 50 species) that have been studied [4].

The existing studies on plant–pollinator and seed disperser–plant interactions also suggest that the nested structure of mutualistic networks reflects a fundamental difference from antagonistic webs, arising from how specialisation is distributed among interacting species [4–6]. In contrast to mutualistic networks, antagonistic networks (e.g., predator– prey, herbivore–plant) tend to be more compartmentalised, i.e., characterised by cohesive groups of interacting species with relatively few interactions among groups [7,8]. Several authors have suggested that nested patterns of asymmetrical specialisation may be more likely to develop in mutualistic interactions because natural selection specifically favours the convergence and complementarity of traits in interacting species [9]. In contrast, antagonistic interactions may favour greater compartmentalisation through the continual coevolution of defences and counterdefences (i.e., evolutionary arm races involving exploitation barriers), which generates greater specificity [9].

The idea that nested asymmetries in specialisation are generated by coevolutionary processes has strong links to the concept of "forbidden links" [5]. Under this hypothesis, the topological properties of mutualistic networks result from nonmatching biological attributes of species, such as phenology or morphology, which prevent the occurrence of certain interactions [8]. However, no work to date has explicitly evaluated the extent to which forbidden links can account for network topology, and detailed information about the biology of species interactions is considered a prerequisite for critical tests [10]. The goal of this study is to bridge this gap by considering the extent to which different ecological mechanisms might account for the observed network patterns. Using a set of simple models, we investigated whether linkage rules aimed at representing the two processes outlined above (trait complementarity and exploitation barriers) can explain the topological properties of plant-pollination networks.

We concentrate on two possible mechanisms. In the first mechanism (complementarity traits), the similarity between the reward that the plant has to offer and the resource that

Academic Editor: Lars Chittka, University of London, United Kingdom

Received May 3, 2006; Accepted December 1, 2006; Published January 23, 2007

**Copyright:** © 2007 Santamaría and Rodríguez-Gironés. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Abbreviations:** C, connectivity; F, number of plant species; L, number of interactions; N, nestedness; N\*, relative nestedness; N<sub>R</sub>, null-model nestedness; P, number of pollinator species; S, network size (number of species); T, temperature

\* To whom correspondence should be addressed. E-mail: luis.santamaria@uib.es

#### Author Summary

Whether they are antagonistic—as between predator and prey—or beneficial—as between pollinator and flower, interactions among all the key species in an ecosystem follow regular patterns. Connectivity (the proportion of possible interactions that are actually realised), for instance, decreases with network size. The "forbidden links" hypothesis proposes that connectivity decreases because interactions are prevented by a mismatch of biological attributes between certain species. Mismatches could arise from the evolution of complementary traits in mutualistic relationships (such as insects preferring to pollinate only flowers of a certain colour) or of traits that prevent exploitation in antagonistic ones (such as a plant growing a long corolla so that insects without a long proboscis cannot reach the nectar reward). We explored the consequences of simple linkage rules based on these two variants on the topology of plant-pollination networks. When compared to data for 37 real plant-pollinator networks, we show that a "mixed" model that combines simple rules from both "complementarity" and "barrier" models best explains the pattern of interactions. This implies, for example, that deterring floral parasites is at least as important as increasing pollination efficiency in the evolution of plant-pollinator networks. Our work emphasises the value of explaining the underlying ecological and evolutionary mechanisms generating such patterns.

the pollinator seeks determines whether species pairs interact. As an example, consider plant and pollinator phenology: a plant will receive visits only of pollinators that are present and active during their flowering period, and plant and pollinator species can become progressively cospecialised by developing increasingly narrower phenological matches [11]. Other examples may include nectar sugar concentration, in which each major pollinator group prefers a specific range of nectar concentrations or sugar compositions [12–15] (although other authors have interpreted nectar dilution as a barrier against bee pollination [16], similar to our second scenario); flower colour, in which flower signals match pollinator perceptual systems (e.g., [17–20], but see also [21,22] for a critical view); and specialised scents [23–25].

In the second scenario (barrier traits), what determines

whether species pairs interact is not their similarity but rather the ability of the pollinator to reach the reward offered by the flowers. Each flower type conceals its reward behind barriers, and only those pollinators whose traits allow them to overcome the barriers have access to the reward. As an example, consider the length of hawkmoths' proboscises and corolla tubes [26,27]: hawkmoths can forage efficiently only at flowers with corolla tubes shorter than their proboscises, and it is mainly these flowers that they visit, as shown by Haber and Frankies [28] for a Costa Rican hawkmoth community. Similar examples, involving proboscis or beak length and corolla-tube depth, have been described for bee-, fly-, and hummingbird-pollinated systems [29–33].

To evaluate whether simple linkage rules lead to mutualistic networks with topological properties similar to those observed in actual networks, we used different sets of rules, derived from the stated mechanisms, to simulate the linkage process. We used complementarity and barrier models based on one, two, or four traits and a mixed model based on two complementarity and two barrier traits. For example, in the model with two complementarity and two barrier traits, each individual (plant or pollinator) in a simulated community was assigned two complementarity and two barrier traits at random, and pairs of plants and pollinators interacted if the barrier traits of the pollinator had higher values than the respective traits of the plants and their complementarity traits were "sufficiently" similar. The complementarity models considered three variants (the narrow, medium, and broad complementarity models), depending on how tight the overlap between the plant and pollinator traits had to be for the species to interact. The topology of simulated communities was then compared with the topology of 37 real-world plant-pollinator networks [3,4,34-36], including three unpublished matrices kindly provided by J. M. Olesen. For conciseness, we only describe the results of the models providing a closer fit to the data.

The assumption that the structure of mutualistic networks results from random interactions among individuals can reproduce certain topological properties of these networks

		-			
Model	Percent of Networks Nested	Percent of Networks Not Nested	Size Range of Nested Networks	Size Range of Nonnested Networks	
	05	r	27.052	22.50	
Real-world data	95	5	27-952	22-50	
Barrier model, I trait	100	0	46-980		
Barrier model, 2 traits	100	0	38–961		
Barrier model, 4 traits	100	0	22-806		
Complementarity model, 1 trait, broad range	100	0	52–980		
Complementarity model, 1 trait, medium range	98.5	1.5	53–980	53–56	
Complementarity model, 1 trait, narrow range	95.5	4.5	52–980	50–66	
Complementarity model, 2 traits, broad range	100	0	47–980		
Complementarity model, 4 traits, broad range	100	0	49–980		
Complementarity model, 2 traits, narrow range	90.5	9.5	25–1,116	22–156	
Complementarity model, 4 traits, narrow range	75	25	29–717	20–103	
Mixed model	89	11	23–714	22–63	
Neutral model (uniform distribution)	100	0	43–1,064		
Neutral model (lognormal distribution)	96	4	20–1,196	23–57	

#### Table 1. Percentage of Modelled Networks that Were Not Significantly Nested

S, network size (number of plant species + number of pollinator species). Sample size: N = 37 for real-world data, N = 200 for all models. doi:10.1371/journal.pbio.0050031.t001

Table 2. R	Relationship	between	Network	Topology	and	Network	Size	(s)
------------	--------------	---------	---------	----------	-----	---------	------	-----

Model	Ν	N <sub>R</sub>	N*	c	L
Real-world data	0.038 imes LnS $+$ 0.74	0.065 imes LnS $+$ 0.52	-0.041 imes LnS $+$ 0.32	$303 imes S^{-0.76}$	$1.08 \times S^{1.10}$
Barrier model, 1 trait	$0.0019 \times LnS + 0.99$	$-0.042 \times LnS + 0.74$	$0.157 \times LnS + 0.115$	$53.7 \times S^{-0.0095}$	$0.26 \times S^{1.81}$
	t = 4.29, p = 6 × 10 <sup>-5</sup>	t = 10.2, p = 2 × 10 <sup>-12</sup>	$t = 20.2, p = 2 \times 10^{-21}$	$t = 11.1, p = 2 \times 10^{-13}$	t = 13.5, p = 6 × 10 <sup>-16</sup>
Barrier model, 2 traits	0.010 × LnS + 0.90	$-0.038 \times LnS + 0.73$	$0.14 \times LnS - 0.049$	$50.9 \times S^{-0.10}$	$0.26 \times S^{1.70}$
	t = 1.18, p = 0.12	t = 9.89, p = 4 × 10 <sup>-12</sup>	$t = 17.5, p = 2 \times 10^{-19}$	t = 8.32, p = 3 × 10 <sup>-10</sup>	$t = 11.0, p = 2 \times 10^{-13}$
Barrier model, 4 traits	0.018 × LnS + 0.88	$0.012 \times LnS + 0.70$	$-0.0022 \times LnS + 0.26$	$59.9 \times S^{-0.30}$	$0.25 \times S^{1.55}$
	t = 0.81, p = 0.21	$t = 4.17, p = 9 \times 10^{-5}$	t = 7.8, p = 10 <sup>-9</sup>	t = 4.46, p = 4 × 10 <sup>-5</sup>	t = 5.96, p = 3 × 10 <sup>-7</sup>
Complementarity model, 1 trait, broad range	-0.072  imes LnS + 0.90	-0.107  imes LnS + 0.94	0.237  imes LnS - 0.81	$23.5 \times S^{-0.0028}$	$0.12 \times S^{1.81}$
Complementarity model,	$t = 5.81, p = 6 \times 10^{-7}$	$t = 11.4, p = 8 \times 10^{-14}$	$t = 11.4, p = 7 \times 10^{-14}$	$t = 13.1, p = 10^{-15}$	$t = 15.9, p = 3 \times 10^{-18}$
1 trait, medium range	-0.033 × LnS + 0.67	-0.072 × LnS + 0.67	$0.327 \times LnS - 0.97$	$41.2 \times S^{0.0061}$	$0.21 \times S^{1.81}$
Complementarity model,	$t = 20.5, p = 9 \times 10^{-22}$	$t = 16.1, p = 3 \times 10^{-18}$	$t = 7.18, p = 9 \times 10^{-9}$	$t = 9.57, p = 10^{-11}$	$t = 11.9, p = 2 \times 10^{-14}$
1 trait, narrow range	0.0087 × LnS + 0.80	-0.060 × LnS + 0.77	0.274 × LnS - 0.27	68.3 × S <sup>0.0057</sup>	$0.35 \times S^{1.81}$
Complementarity model,	$t = 13.9, p = 2 \times 10^{-16}$	$t = 10.5, p = 7 \times 10^{-13}$	$t = 5.74, p = 7 \times 10^{-7}$	$t = 5.89, p = 5 \times 10^{-7}$	$t = 7.52, p = 3 \times 10^{-9}$
2 traits, broad range	-0.022 × LnS + 0.79	-0.062 × LnS + 0.69	0.26 × LnS - 0.46	1.01 × S <sup>0.013</sup>	6.30 × S <sup>1.84</sup>
Complementarity model,	$t = 12.4, p = 8 \times 10^{-15}$	$t = 14.6, p = 5 \times 10^{-17}$	$t = 11.8, p = 3 \times 10^{-14}$	$t = 10.5, p = 7 \times 10^{-13}$	$t = 12.4, p = 6 \times 10^{-15}$
4 traits, broad range	-0.032 × LnS + 0.83	-0.079 × LnS + 0.83	$0.23 \times LnS - 0.62$	3.38 × S <sup>-0.025</sup>	5.88 × S <sup>1.77</sup>
Complementarity model,	$t = 11.6, p = 5 \times 10^{-14}$	$t = 11.0, p = 2 \times 10^{-13}$	$t = 7.23, p = 8 \times 10^{-9}$	$t = 6.5, p = 8 \times 10^{-8}$	$t = 8.71, p = 10^{-10}$
2 traits, parrow range	-0.0098 × LnS + 0.91	-0.014 × LnS + 0.87	0.0068 × LnS + 0.038	0.82 × S <sup>-0.20</sup>	4.86 × S <sup>1.58</sup>
Complementarity model,	$t = 4.81, p = 1 \times 10^{-5}$	t = 3.87, p = 0.00022	t = 2.55, p = 0.0076	$t = 4.49, p = 3 \times 10^{-5}$	<b>t</b> = <b>4.10, p</b> = <b>0.00011</b>
4 traits, narrow range	0.048 × LnS + 0.70	$0.049 \times LnS + 0.67$	−0.0021 × LnS + 0.036	0.47 × S <sup>-0.75</sup>	3.26 × S <sup>1.18</sup>
Mixed model	t = 1.03, p = 0.16	$t = 4.56, p = 3 \times 10^{-5}$	$t = 5.03, p = 7 \times 10^{-6}$	$t = 17, p = 4 \times 10^{-19}$	$t = 14.2, p = 10^{-16}$
	$0.030 \times LnS + 0.79$	0.035 × LnS + 0.70	-0.0091 × LnS + 0.12	$51.8 \times S^{-0.49}$	3.95 × S <sup>1.37</sup>
	t = 0.64, p = 0.26	t = 1.77, p = 0.042	t = 1.47, p = 0.07	$t = 5.21, p = 4 \times 10^{-6}$	$t = 5.81, p = 5 × 10^{-7}$
Neutral model (uniform distribution)	$-0.0042 \times LnS + 0.76$	$-0.062 \times LnS + 0.77$	$0.23 \times LnS - 0.53$	$35.72 \times S^{-0.050}$	$0.18 \times S^{1.76}$
Neutral model	$t = 10.2, p = 2 \times 10^{-12}$	$t = 11.4, p = 7 \times 10^{-14}$	$t = 8.74, p = 10^{-10}$	$t = 7.10, p = 10^{-8}$	$t = 9.24, p = 2 \times 10^{-11}$
(lognormal distribution)	0.026 × LnS + 0.81	0.045 × LnS + 0.66	-0.025 × LnS + 0.19	108 × S <sup>-0.64</sup>	0.48 × S <sup>1.21</sup>
	t = 0.12, p = 0.45	<i>t</i> = 1.25, <i>p</i> = 0.11	t = 0.93, p = 0.18	t = 4.73, $p =$ 2 $ imes$ 10 <sup>-5</sup>	t = 4.94, $p =$ 9 $ imes$ 10 <sup>-6</sup>

Paired t-test values and significance levels compare each model to real-world data (df = 36), by comparing the absolute value of the residuals of real-world data to an empirical fit (shown in the first row) with the deviations of the real-world data points from fits obtained from the modelled data sets. For each variable, models that did not differ significantly from the empirical fit to real-world data (i.e., those with nonsignificant t-tests, after applying sequential Bonferroni corrections) are marked in bold. When all models differed significantly from the empirical fit, the model with the lowest residual sum of squares plus all models that did not show a significantly worse fit (i.e., significantly larger residuals when compared with paired t-tests, after applying sequential Bonferroni corrections) are marked in bold. Under the tests, after applying sequential bonferroni corrections are marked in bold. When all models differed significant t-tests, after applying sequential Bonferroni corrections) are marked in bold. When all models differed significant t-tests, after applying sequential Bonferroni corrections are marked in bold. When all models differed significant t-tests, after applying sequential Bonferroni corrections) are marked in bold. When all models differed significant t-tests, after applying sequential Bonferroni corrections are marked in bold italics. doi:10.1371/journal.pbio.0050031.t002

[6,10]. To complement our work on mechanistic linkage rules, we used two formulations of the neutrality hypothesis of Vázquez and Aizén [6] to simulate plant-pollinator networks. In these models, the probability that a plant-pollinator pair interacts is proportional to their relative abundance, and relative abundances were drawn from either a uniform ("uniform neutral model") or a lognormal ("lognormal neutral model") probability distribution [37,38].

# Results

Most (95%) real-world networks were significantly nested, all exceptions being networks of small size (network size [S] < 50). Most modelled networks were also highly nested, although a few models showed a small proportion (up to 11% for the mixed model) of nonnested networks for small network sizes (S < 160 and, in most cases, S < 70; Table 1).

The scaling properties of networks of increasing species richness (S) differed substantially from those of real-world

pollination networks for all one-trait models, most two- and four-trait models, and the uniform neutral model (Table 2; Figures 1 and 2). Only the most restrictive complementarity (two and four traits, narrow range), the four-trait barrier model, the mixed model, and the lognormal-neutral model approached the trends observed in real-world data, and the latter two provided the best fits to most variables.

In real-world networks, the connectivity [C] decreases as a power of S (Figure 1) [3]. While five models (two- and fourtrait barrier, four-trait narrow complementarity, mixed, and lognormal-neutral models) approached this behaviour, all of them failed to predict real-world values across the complete range of network sizes and intersected the empirical fit at the lower, medium, and upper part, respectively, of the data range (Figure 1).

The number of interactions (L) increased as a power of S in all models. The exponent of this relationship was greater than the one describing real-world data for all models, and the



Figure 1. Connectivity of Networks of Increasing Size (S) for Real-World and Modelled Data Matrices

Circles indicate real-world data (N = 37), the continuous black line indicates an empirical fit to real-world data, and the broken black line is the result of lognormal-neutral models. Colour lines indicate the fits to modelled data sets: narrow-range complementarity models are red, broad-range complementarity models are red, but, and mixed models are green. Dotted lines indicate one-trait models, broken lines indicate two-trait models, and continuous lines indicate four-trait models. doi:10.1371/journal.pbio.0050031.g001

four-trait barrier, two-trait narrow complementarity, and lognormal-neutral models provided the closest fits (Table 2).

In real-world pollination networks, both the nestedness of the networks (N) and the nestedness of random matrices with similar size and average C (null-model nestedness  $[N_R]$ ) increased as the logarithm of S (Table 2; Figure 2). This simultaneous increase of N and  $N_R$  results in a logarithmic decrease of the relative nestedness  $[N^*]$ . Five models (two-and four-trait barrier, two-trait narrow complementarity, mixed, and lognormal-neutral models) reproduced remarkably well the logarithmic increase in N with network size, but only two of them (mixed and lognormal-neutral models) performed well in predicting the nestedness of random matrices ( $N_R$ ) and, therefore, the N\* (Figure 2).

Bascompte et al. [4] divided their mutualistic networks into two groups, according to whether their number of interactions (L) was greater or lesser than expected for their size (positive or negative residuals, respectively, in the regression between L and S). The N\* in networks having positive residuals was greater than that in networks with negative residuals (F = 6.59, df = 1, 50, p = 0.013) [4]. This analysis pooled pollination and seed-dispersal networks, and the results did not reach significance when plant-pollinator networks (of which there were 25) were analysed on their own. However, the addition of 12 networks to the data set used here already resulted in a significant difference (t = 3.51, p = 0.0015, n = 37). Running this analysis on the networks generated by the models, we observed that four models (fourtrait barrier, two-trait narrow-complementarity, mixed, and lognormal neutral models) produced comparable results (Figure 3).

Because N\* decreases with  $N_R$  and, in the range of interest (C < 50%),  $N_R$  decreases with C [39], the increased N\* of positive L-on-S residuals implies that, as the number of



Figure 2. Nestedness of Networks of Increasing Size (S) for Real-World and Modelled Data Matrices

Circles indicate real-world data (N = 37), the continuous black line indicates an empirical fit to real-world data, and the broken black line is the result of lognormal-neutral models. Colour lines indicate the fits to modelled data sets: narrow-range complementarity models are red, broad-range complementarity models are magenta, barrier models are blue, and mixed models are green. Dotted lines indicate one-trait models, broken lines indicate two-trait models, and continuous lines indicate four-trait models. N<sub>R</sub> (random matrices). N\*, relative nestedness (N/NR)

doi:10.1371/journal.pbio.0050031.g002

connections of mutualistic networks increases, the nestedness of real-world networks does not decrease as fast as it would in a random matrix. A multiple regression analysis showed that the trends described for real-world data (decreased N and  $N_R$  and increased N\* with increased L, all corrected for the effect of S; Table 3) were shown by the mixed, lognormal-neutral, and four-trait narrow-complementarity models.



Figure 3. Relationship between Network C and N\* for Real-World and Modelled Data Matrices

Insets show the N\* (average  $\pm$  SE) of highly and lowly connected communities (i.e., those with positive and negative residuals in the fits of the number of interactions on species richness displayed in the larger panels). Narrow-range complementarity models are red, broad-range complementarity models are magenta, barrier models are blue, and mixed models are green. Dotted lines indicate one-trait models, broken lines indicate two-trait models, and continuous lines indicate four-trait models. Circles indicate real-world data (N = 37), the continuous black line indicates an empirical fit to real-world data, and the broken black line is the result of lognormal-neutral models. Asterisks indicate significant differences between highly and lowly connected categories (i.e., those with positive and negative residuals, respectively). \*p < 0.05, \*p < 0.01, \*\*\*p < 0.001, NS p > 0.05. doi:10.1371/journal.pbio.0050031.g003

### Discussion

Recent work has greatly improved our knowledge of the patterns and scaling laws that characterise plant-pollinator networks, and the current challenge is to understand the ecological and evolutionary processes that underlie these regularities. Our work represents a first step in this direction.

Given the excellent performance of the lognormal neutral model in predicting network characteristics, what is the use of exploring more complex linkage rules? The main purpose of null models is to remind us that showing that a given model fits the data well is not necessarily a demonstration that the model is "correct": more than one mechanism can produce any given pattern [40]. When two alternatives can explain a

certain phenomenon, parsimony demands that we provisionally accept the simpler one, and null models are meant to represent the simplest explanation that can be offered. So, if a null model can be used to explain a pattern, there is no good reason to search for a more complex explanation. There are three reasons, however, why the lognormal neutral model can be rejected as a most-parsimonious explanation of network topology: (1) Assuming random interactions, in itself, is not sufficient to reproduce network topology: the uniform null model provided a very poor fit to the data, and in order to fit the model we had to assume a lognormal distribution of abundances. While the distribution of species abundances in many communities is well described by a lognormal distribution [41], this is an empirical fit. The interpretation of this empirical fit is problematic because it is not clear whether generalist species are generalist because they are more abundant, or they are more abundant because, being generalists, they have access to more resources. (2) The neutral model assumes that species abundance determines the frequency of interactions. This assumption has been recently evaluated by Dupont et al. [36], who found a significant correlation between abundance and generalisation level of the pollinators, but not the plants, in an alpine community. Ollerton et al. [42], on the other hand, reported an association between interaction frequency and generalisation level but found no significant correlation between the relative abundance of insects in the community and their visitation rate to asclepiad plant species. (3) The neutral model assumes that network structure results from random interactions between species and "most phenotypic characteristics of interacting species may be irrelevant in determining broad patterns of interspecific interactions" (e.g., degree distribution) [10]. This assumption is at odds with all reported empirical data, which show that phenotypic traits often prevent the interaction between specific pairs of plants and pollinators [8,33]. For example, Jordano et al. [8] show how 42% of 65% nonrecorded interactions in a plant-hummingbird subnetwork can be attributed to phenological or phenotypic mismatches between the plant species and the pollinator, indicating that a sizeable fraction of the interactions are "forbidden" and thus individuals cannot interact at random. Given that two key assumptions of the neutral model are not supported by empirical data and that its causal interpretation is problematic, and bearing in mind that more than one mechanism can produce any given pattern, we believe that an examination of alternative models based on phenotypic traits is granted.

Let us now turn to the ecological mechanisms (linkage rules) that we have considered. Of these, only the multipletrait models were compatible with the data, and the combination of both barrier and complementarity rules fitted them best. Although we have tried a number of other mechanistic models, such as stochastic versions of the complementarity and barrier models, and models that combined a lower amount of complementarity and barrier traits (not dealt with in depth for the sake of conciseness), we have been unable to produce any version that fitted the data better than the mixed model presented here.

Although the optimal fit of the four-trait mixed model does not necessarily imply that it provides a realistic description of the linkage rules responsible for network assembly in the real world [40], we have shown that simple ecological processes may well lie behind the complexity of these large networks,

#### Table 3. Relationship between Network N, S, and C

Model	Variable Fitted Equation		Partial Correlations		
			LnS	LnL	
	Ν		0.40**	0.26*	
Real-world data	IN N	$0.75 \pm 0.11 \times LnS = 0.067 \times LnL$	0.49**	-0.36"	
	IN <sub>R</sub>	$0.52 + 0.22 \times LnS - 0.14 \times LnL$	0.76***	-0.64***	
Parriar model 1 trait	IN."	$0.31 - 0.18 \times LH3 + 0.12 \times LHL$	-0.76 <sup>NS</sup>	0.06	
bamer model, i trait	IN N	$0.99 - 0.003 \times LHS - 0.003 \times LHL$	-0.06	0.14"	
	INR N*	$0.01 + 0.13 \times LHS = 0.093 \times LHL$	0.00	-0.71	
Parriar model 2 traits	N	$0.46 - 0.53 \times LII5 + 0.27 \times LIIL$	-0.55	-0.70	
barrier model, 2 traits	N.	$0.00 \pm 0.070 \times 115 = 0.055 \times 1112$	0.50	-0.31	
	INR N*	$0.43 + 0.32 \times LIIS - 0.21 \times LIIL$	0.90	-0.92***	
Barrier model 4 traits	N	$0.00 - 0.08 \times LIIS + 0.00 \times LIIL$	-0.91	0.93	
	N.	$0.79 \pm 0.11 \times Ens = 0.000 \times Enc$	0.00	-0.45	
	N*	$0.30 \pm 0.30 \times LnS \pm 0.32 \times LnL$	0.95	-0.95	
Complementarity model 1 trait broad range	N	$0.77 = 0.30 \times LnS + 0.16 \times LnL$	-0.00	0.00	
complementanty model, i trait, broad range	N.	$0.37 = 0.23 \times 105 = 0.062 \times 101$	-0.42	0.45	
	INR N*	$0.70 \pm 0.003 \times \text{LHS} = 0.002 \times \text{LHL}$	0.07	-0.10	
Complementarity model 1 trait modium range	N	$-0.16 - 0.31 \times LI13 + 0.32 \times LI12$	-0.22	0.59 0.06 <sup>NS</sup>	
Complementarity model, i trait, medium range	IN N	$0.72 - 0.083 \times Lins + 0.029 \times Linc$	-0.09	0.00	
	INR N*	$0.43 + 0.20 \times LHS = 0.13 \times LHL$	0.45	-0.57***	
Complementarity and del 1 tarity any manage	IN"	$-0.23 - 0.53 \times LnS + 0.47 \times LnL$	-0.39***	0.57***	
Complementarity model, 1 trait, narrow range	N	$0.38 + 0.37 \times LnS - 0.24 \times LnL$	0.00***	-0.50***	
	N <sub>R</sub>	$0.32 + 0.42 \times LnS = 0.29 \times LnL$	0.88***	-0.92***	
Complementarity model 2 tority based are as	IN"	$-0.59 + 0.056 \times LnS + 0.10 \times LnL$	0.03	0.11 <sup>NS</sup>	
Complementarity model, 2 traits, broad range	N	$0.72 + 0.050 \times LnS - 0.039 \times LnL$	0.07***	-0.11***	
	N <sub>R</sub>	$0.45 + 0.21 \times LnS - 0.15 \times LnL$	0.23***	-0.31***	
Complementarity model 4 traits bread range	N*	$0.13 - 0.43 \times LnS + 0.37 \times LnL$	-0.61***	0.77***	
Complementarity model, 4 traits, broad range	N	$0.67 + 0.13 \times LnS - 0.089 \times LnL$	0.27***	-0.33***	
	N <sub>R</sub>	$0.38 + 0.36 \times LnS - 0.25 \times LnL$	0.87***	-0.91***	
	N*	$0.39 - 0.74 \times LnS + 0.55 \times LnL$	-0.71***	0.80	
Complementarity model, 2 traits, narrow range	N	$0.50 + 0.29 \times LnS - 0.19 \times LnL$	0.74***	-0.75***	
	N <sub>R</sub>	0.45 + 0.29 × LnS – 0.19 × LnL	0.93***	- <b>0.94</b> ***	
	N*	$0.094 - 0.034 \times LnS + 0.026 \times LnL$	-0.11	0.13	
Complementarity model, 4 traits, narrow range	N	0.66 + 0.086 × LnS – 0.032 × LnL	0.18**	-0.08	
	N <sub>R</sub>	0.56 + 0.16  imes LnS - 0.10  imes LnL	0.70***	-0.57***	
	N*	0.14 $-$ 0.10 $ imes$ LnS $+$ 0.084 $ imes$ LnL	<b>-0.22**</b>	0.22**	
Mixed model	N	0.65 $+$ 0.15 $ imes$ LnS $-$ 0.091 $ imes$ LnL	0.49***	- <b>0.42</b> ***	
	N <sub>R</sub>	0.52 $+$ 0.20 $ imes$ LnS $-$ 0.12 $ imes$ LnL	0.82***	- <b>0.76</b> ***	
	N*	0.18 $-$ 0.067 $ imes$ LnS $+$ 0.042 $ imes$ LnL	<b>-0.24</b> **	0.21**	
Neutral model (uniform distribution)	Ν	0.56 $+$ 0.20 $ imes$ LnS $-$ 0.11 $ imes$ LnL	0.43***	-0.44***	
	N <sub>R</sub>	0.33 + 0.39 $ imes$ LnS - 0.26 $ imes$ LnL	0.90***	-0.92***	
	N*	0.40 $-$ 0.071 $ imes$ LnS $+$ 0.53 $ imes$ LnL	-0.77**	0.85***	
Neutral model (Lnnormal distribution)	N	0.76 $+$ 0.10 $ imes$ LnS $-$ 0.061 $ imes$ LnL	0.79***	- <b>0.70</b> ***	
	N <sub>R</sub>	0.58 $+$ 0.17 $ imes$ LnS $-$ 0.10 $ imes$ LnL	0.94***	- <b>0.90</b> ***	
	N*	0.23 $-$ 0.095 $ imes$ LnS $+$ 0.058 $ imes$ LnL	- <b>0.78</b> ***	0.69***	

Figures indicate partial correlations and significance levels for multiple regression fits that included both independent variables. For each dependent variable, the best models (i.e., those did not differ significantly from the empirical fit to real-world data, compared using paired *t*-tests with sequential Bonferroni correction, df = 36) are marked in bold. \*p < 0.05, \*\*p < 0.01, \*\*\*p < 0.01, \*\*\*p > 0.05.

doi:10.1371/journal.pbio.0050031.t003

and this result should encourage us to search for linkage rules in the field. A minimum amount of complexity is nevertheless required to explain real-world networks, since a combination of at least four traits was necessary to reproduce the patterns observed there.

The poor fit of the complementarity models to the data shows that the complementarity rule, in itself, cannot be the linkage rule we are looking for. It should be noted that this result can be taken at face value: a model providing a poor fit to the data can be dismissed regardless of whether a better model is known. The complementary and barrier traits had contrasting effects on network characteristics. Plant–pollinator pairs specialised on each other predominate with complementarity models, leading to highly connected networks of low nestedness. To approach the nestedness of realworld networks, we must impose very restrictive conditions (narrow ranges with several traits), and under these conditions, networks are very sensitive to random effects (high values of  $N_R$ ) and show excessively low C. With barrier models, on the other hand, specialised plants interact with pollinators that have access to very diverse resources, producing highly nested networks. In mixed models, which provide the best fits, complementary traits relax the trend to excessive nestedness of barrier models, and barrier models relax the too-low connectance and the high dependence of random effects of complementary traits.

The demonstration that the complementary rule alone is unable to produce realistic network topologies has important evolutionary implications. It suggests that nested patterns of asymmetrical specialisation observed in mutualistic interactions do not arise because natural selection on mutualisms specifically favours the convergence and complementarity of traits in interacting species. Plant-pollinator coevolution may be the result of selection for plant traits that enhance visitation rates by the most efficient pollinator and/or pollination efficiency by the most common pollinator [1,17,43], but floral evolution might also represent a compromise between attraction and defence [44-48]. Although this latter hypothesis has been questioned because deterrent traits will interfere with nectar exploitation by efficient pollinators [49], recent models incorporating foraging decisions have shown that the evolution of pollinator-deterring traits is possible, provided that deterring parasites (i.e., floral visitors that, by reducing pollen availability to more-efficient pollinators, decrease the plant's fitness) is sufficiently beneficial [50]. Facilitation of most-efficient pollinators predicts an increasing narrowing of the ecological range of plants and pollinators as a consequence of coevolution. In such a scenario, evolution leads to an increasingly tighter match of the coevolved morphological structures (and/or functional and behavioural traits) of plants and animals: the complementarity rule. In contrast, parasite deterrence leads to the evolution of floral barriers and pollinator structures allowing them to overcome plant defences: the barrier rule. The inability of the complementarity rule to mimic the topology of real-world networks thus suggests that the most-efficientpollinator principle is not the main (or the exclusive) driving force behind the evolution of floral and pollinator traits. The optimal performance of the mixed model indeed suggests that both the most-efficient-pollinator principle and parasite deterrence operate simultaneously as evolutionary forces in natural communities.

Barrier traits also provide an answer to Vázquez's main criticism of Jordano's forbidden interactions: "forbidden links resulting from phenological or morphological constrains are equally likely to affect any species, not just the most connected ones, and it is unclear whether this assembly constraint would necessarily lead to a decay in the tail of the degree distribution" [10]. By its very nature, barrier traits affect specialist and generalist, plant and insect species asymmetrically. While large trait values result in ecological specialisation for plant species (i.e., few insects have trait values large enough to access their resources), they result in ecological generalisation for insect species (i.e., insects with large trait values have access to many plant species). The asymmetrical effect of large trait values on plants and insects automatically generates nested patterns in the interaction networks. More important, competition for floral resources results in niche segregation that favours the exploitation of specialised (i.e., large trait) plants by generalist (i.e., large trait) insects, and vice versa [50], thereby imposing an upper limit to the number of interactions that a generalist can sustain at any given time.

The nested pattern seems to be a pervasive feature of mutualistic networks and has also been found in seeddispersal networks and, more recently, in plant-ant [9] and fish-anemone [51] communities. Our results suggest that these systems may also be characterised by the coexistence of both complementarity and barrier traits. Examples of both types of traits for seed-dispersal networks may respectively include fruiting phenology and disperser availability, on the one hand, and fruit size and gape width, on the other [52]. Nevertheless, it must be noted that our work studies, other than network nestedness, the relationships between network size and various topological properties (L, C, N, N<sub>R</sub>, and N\*). In order to explore the applicability of complementarity and barrier linkage rules to other mutualistic systems, we would need to know how the topological properties of these networks scale with size.

Our study represents a first effort to go beyond the description of network topology and into the analysis of the ecological and evolutionary processes behind it, which may complement (and aid the interpretation of) other attempts based on empirical studies [8,36,42]. The results presented in this paper stress the utility of mechanistic, phenotype-based approaches to community-level questions. To elucidate the actual ecological processes responsible for the assembly of plant–pollinator networks, we need detailed quantitative descriptions of the plant–pollinator networks, the relationship between traits of interacting plant–pollinator pairs, and the implications of phenotypic traits for plant and pollinator fitness.

#### **Materials and Methods**

We restrict our analysis and discussion to qualitative networks. Plant-pollinator networks are characterised by their interaction matrix, in which rows represent pollinator species and columns represent plant species. The ij cell of the matrix is set to "1" if pollinator species i interacts with plant species j, and to "0" otherwise.

To evaluate whether simple linkage rules lead to mutualistic networks with the same topology as the networks actually observed, we used different sets of rules to simulate the linkage process and compared the topology of simulated communities with the topology of 37 real-world plant-pollinator networks [3,4,34–36], plus three unpublished matrices kindly provided by J. M. Olesen.

For each model, we simulated 200 communities of different sizes by selecting a random number of potential plant species (F; in the range of 10 to 160, uniform distribution), letting the number of potential pollinators be  $P = 45.573 - 0.8082 \times F + 0.047 \times F^2$  (this regression was derived from the real-world communities considered in the study) and assigning traits at random to plants and pollinators to determine which pairs interacted (as determined below for each particular model). We then removed all plants and pollinators that had no interactions, arriving at F1 and P1 interacting plants and pollinating species, respectively. Relationships between network topology and community size are all based on the value of  $S_1 = F_1 + P_1$ . Network topology was described by the total L, C, N, and N\* [4,5]. A mutualistic network is perfectly nested if each plant species is visited by a subset of the pollinators visiting more-generalist plants [4,5], the "temper-(T) of the network is a normalised measure of the network's ature' deviation from perfect nestedness [4,5] and ranges from 0 to 100, and the level of nestedness is defined as N = (100 - T)/100 and ranges from 0 to 1 [4]. For each community that we simulated, we used 100 random matrices to assess the extent to which the level of nestedness of the hypothetical community deviated from random expectations and to calculate N\*: N\* = ( $\dot{N} - N_R$ )/N<sub>R</sub>, where N<sub>R</sub> is the average nestedness of the random replicates ("null-model nestedness"). The random replicates were generated using the most conservative null model (null model 2) proposed by Bascompte et al. [4].

Interaction matrices were generated with a C++ program, available upon request, as detailed below for each model. All "random" numbers were generated with the function "ran1" from [53]. The program calculates nestedness using BINMATNEST (available at http://www.eeza.csic.es/eeza/personales/rgirones.aspx [39]), an improved version of the Nestedness Temperature Calculator [54,55].

**Single-trait complementarity model.** The single-trait complementarity model assumes that plants and pollinators can be described by a single trait. Each species is characterised by a mean trait value and a range of variability, and a pair of species will interact if their traits overlap. Let  $V_i$  and  $W_j$  be the central trait value for pollinator species *i* and plant species *j*, respectively, and let  $\delta V_i$  and  $\delta W_j$  be their ranges of variability. Then, the value of the interaction matrix corresponding to this pair of species,  $I_{ij}$ , will be

$$I_{ij} = 1 \text{ if } |V_i - W_j| < 0.5 \times (\delta V_i + \delta W_j)$$

$$\tag{1}$$

and

$$I_{ii} = 0$$
 otherwise. (2)

We considered three scenarios (hereafter referred to as broad-, medium-, and narrow-range complementarity models, respectively) that differed in the average value of  $\delta V_i$  and  $\delta W_j$ . In all cases, the  $V_i$ and  $W_j$  were independent random variates with uniform distribution in the interval (0, 1). The  $\delta V_i$  and  $\delta W_j$  on the other hand, were random variates with uniform distributions in the intervals (0, 1), (0, 0.5), and (0, 0.25), for the broad-, medium-, and narrow-range complementarity models, respectively.

For many traits, there is a correlation between average value and variability (J. M. Olesen, unpublished data). Including this correlation in the model lead to identical results to the ones with uncorrelated mean and variability, and the results will not be discussed further.

**Two- and four-trait complementarity models.** In most, if not all, real communities, flowers differ along several dimensions. The twoand four-trait complementarity models consider the possibility that pollinators must fit several floral traits in order to reach the reward offered by flowers. The flowers of the *j*th plant species are thus characterised by *N* central trait values and ranges of variability (N=2or 4 for the two- and four-trait complementarity models, respectively),  $W_j^k$  and  $\delta W_j^k$ , with k = 1, ..., N. Similarly, the pollinators of the *i*th species are characterised by central trait values  $V_i^k$  and ranges of variability  $\delta V_i^k$ , with k = 1, ..., N. Central trait values for plants and pollinators were independent random variates with uniform distribution in (0, 1), and ranges of variability were independent random variates with uniform distribution in the range (0, 1), (0, 0.5), or (0, 0.25) for the broad-, medium-, and narrow-range complementarity models, respectively. The interaction matrix is given by

$$I_{ij} = 1 \text{ if } | < V_i >^k - < W_j >^k | < 0.5 \times (\delta V_i^k + \delta W_j^k) \text{ for all } k = 1, \dots N$$
(3)

and

$$I_{ii} = 0$$
 otherwise. (4)

**One-trait barrier model.** Like the complementarity model, the onetrait barrier model assumes that flowers and pollinators can be described by a single trait. We ignore variability in this case and simply assume that pollinators of species *i* will visit flowers of species *j* if their trait,  $V_i$  is greater than the barrier of the plant,  $W_j$  (assuming that pollinators of species *i* will visit flowers of species *j* if  $V_i < W_j$  is mathematically equivalent and leads to exactly the same results). As a result, the interaction matrix takes the values

$$I_{ii} = 1 \text{ if } V_i > W_i \tag{5}$$

and

$$I_{ii} = 0$$
 otherwise. (6)

The  $V_i$  and  $W_j$  were independent random variates with uniform distributions in the interval (0, 1).

**Two- and four-trait barrier models.** The two- and four-trait barrier models consider the possibility that pollinators must overcome several barriers in order to reach the reward offered by flowers. The flowers of the *j*th plant species are thus characterised by *N* barriers (N = 2 or *t* for the two- and four-trait barrier models, respectively),  $W_j^k$ , with k = 1, ..., N, and the pollinators of the *i*th species are characterised by traits  $V_i^k$ , with k = 1, ..., N, where the  $V_i^k$  and  $W_j^k$  are independent random variates with uniform distribution in (0, 1). The interaction matrix is given by

 $I_{ij} = 0$  otherwise.

flowers and pollinators are likely to combine both complementarity

and barrier types of traits (J. M. Olesen, M. Price, and N. Waser

unpublished data). The mixed models combine complementarity and

barrier types of traits (two traits each). For this purpose, we selected

the barrier model that produced the best fit to real-world data: the

(k = 1, 2), and two central trait values,  $V_i^k$  (k = 3, 4). All variables are

independent random variates with uniform distribution and ranges

The flowers of the *j*th plant species are thus characterised by two barrier traits,  $W_j^k$ , two ranges of variability  $\delta W_j^k$  (k = 1, 2), and two central trait values,  $W_j^k$  (k = 3, 4). The pollinators of the *i*th species are characterised by two barrier traits,  $V_i^k$ , two ranges of variability  $\delta V_i^k$ 

Mixed models. In real communities, the interactions between

$$I_{ij} = 1 \text{ if } V_i^k > W_i^k \text{ for all } k = 1, \dots N$$

$$\tag{7}$$

and

Linkage Rules for Pollination Networks

 $(0,\,1)$  for the barrier traits and central trait values and  $(0,\,0.25)$  for the ranges of variability. The interaction matrix is given by

$$I_{ij} = 1 \text{ if } |V_i^k - W_j^k| < 0.5 \times (\delta V_i^k + \delta W_j^k), \text{ for } k = 1 \text{ and } 2 \text{ and}$$
  
$$V_i^k > W_j^k, \text{ for } k = 3 \text{ and } 4$$
(9)

and

$$I_{ij} = 0$$
 otherwise. (10)

**Neutral model.** The neutral models assume that interactions between plants and pollinators are determined by the relative abundance of the species. The *j*th plant species is characterised by its relative abundance,  $W_p$  and the pollinators of the *i*th species by their relative abundance,  $V_i$  where the  $V_i$  and  $W_j$  are independent random variates with uniform distribution in (0, 1). The interaction matrix is given by

$$I_{ij} = 1$$
 with probability  $V_i \times W_j$  (11)

and

$$I_{ij} = 0$$
 otherwise. (12)

**Lognormal neutral model.** The lognormal neutral model was similar to the neutral model, except that the  $V_i$  and  $W_j$  are now independent random variates with lognormal distribution. The interaction matrix is given by

$$U_{ij} = 1$$
 with probability  $V_i \times W_j / \max[V_i \times W_j]$  (13)

and

$$I_{ij} = 0$$
 otherwise. (14)

We studied 200 communities generated with the algorithm specified above and used the same parameters to analyse the topology of the networks generated.

Comparisons among models. Models were compared on the basis of their fit to real-world data. We first fitted separate regression lines (logarithmic for N,  $N_R$ , and N\*, power fit for C and I) to each data set generated by the various model (model fits). For each model, we then calculated the deviation of the real-world data points from the model fit (i.e., the absolute value of the difference between each real-world data point and the expected value, calculated using the model fit) and compared these deviations with the absolute value of the residuals of the real-world data points to a regression line fitted through these points (same type of regression as above: logarithmic for N, N<sub>R</sub>, and N\*, power fit for C and I). Deviations from model fits and residuals were compared using paired t-tests with sequential Bonferroni correction. We used paired t-tests because we compared pairs of distances for each data point: the deviation from the model fit versus the absolute value of the residual from the empirical fit. We considered as best models all those that did not differ significantly from the empirical fit (i.e., those which do not perform significantly worse than an empirical fit). Whenever all models performed significantly worse than the empirical fits (i.e., for C and I), we performed multiple comparisons among all models using paired ttests (similar to above but comparing for each real-world data point the deviation from one model fit versus the deviation from the other model fit) with sequential Bonferroni correction. We considered as best models those with the best fit to real-world data (i.e., the lowest sum of squared deviations) plus all those that did not perform significantly worse (i.e., those that did not show significantly higher residuals) than them.

### Acknowledgments

J. Olesen provided the real-world data matrices analysed here. We thank J. Bascompte, J. Olesen, J. Ollerton, M. Price, D. Vazquez, N. Waser, and one anonymous referee for comments and suggestions that clarified our thoughts and improved previous drafts of this manuscript.

Author contributions. LS and MARG conceived and designed the experiments, ran model simulations, and wrote the paper. LS analyzed the data. MARG wrote the models.

Funding. Funding by the Spanish Ministry of Science and Technology (project INVASRED, REN2003-06962) is also acknowledged.

**Competing interests.** The authors have declared that no competing interests exist.

narrow-complementarity model.

(8)

#### References

- Waser NM, Chittka L, Price MV, Williams NM, Ollerton J (1996) Generalization in pollination systems, and why it matters. Ecology 77: 1043–1060.
- Jordano P (1987) Patterns of mutualistic interactions in pollination and seed dispersal: Connectance, dependence asymmetries, and coevolution. Am Nat 129: 657-677.
- Olesen JM, Jordano P (2002) Geographic patterns in plant-pollinator mutualistic networks. Ecology 83: 2416–2424.
- Bascompte J, Jordano P, Melián CJ, Olesen JM (2003) The nested assembly of plant-animal mutualistic networks. Proc Natl Acad Sci U S A 100: 9383– 9387.
- Jordano P, Bascompte J, Olesen JM (2003) Invariant properties in coevolutionary networks of plant-animal interactions. Ecol Lett 6: 69–81.
- Vazquez DP, Aizen MA (2004) Null model analyses of specialization in plant-pollinator interactions. Ecology 85: 1251–1257.
- Prado PI, Lewinsohn TM (2004) Compartments in insect-plant associations and their consequences for community structure. J Anim Ecol 73: 1168– 1178.
- Jordano P, Bascompte J, Olesen JM (2006) The ecological consequences of complex topology and nested structure in pollination webs. In: Waser NM, Ollerton J, editors. Plant-pollinator interactions: From specialization to generalization. London: The University of Chicago Press. pp. 173–199.
- Guimarães PR, Rico-Gray V, Furtado dos Reis S, Thompson JN (2006) Asymmetries in specialization in ant-plant mutualistic networks. Proc R Soc Lond B 273: 2041–2047.
- Vázquez DP (2005) Degree distribution in plant-animal mutualistic networks: Forbidden links or random interactions. Oikos 108: 21–426.
- Feinsinger P (1987) Effects of plant species on each other's pollination: Is community structure influenced? Trends Ecol Evol 2: 123–126.
- Proctor M, Yeo P, Lack A (1996) The natural history of pollination. Portland (Oregon): Timber Press. 487 p.
- Baker HG, Baker I, Hodger SA (1998) Sugar composition of nectars and fruits consumed by birds and bats in the tropics and subtropics. Biotropica 30: 559–586.
- Perret M, Chautems A, Spchiger R, Peixoto M, Savolaien V (2001) Nectar sugar composition in relation to pollination syndromes in Sinningieae (Gesneriaceae). Ann Bot 87: 267–273.
- Nicolson SW, Fleming PA (2003) Nectar as food for birds: The physiological consequences of drinking dilute sugar solutions. Plant Syst Evol 238: 139– 153.
- Bolten AB, Feisinger P (1978) Why do hummingbird flowers secrete dilute nectar? Biotropica 10: 307–309.
- Schemske DW, Bradshaw HD (1999) Pollinator preference and the evolution of floral traits in monkey flowers (*Mimulus*). Proc Natl Acad Sci U S A 96: 11910–11915.
- Briscoe AD, Chittka L (2001) The evolution of color vision in insects. Annu Rev Entomol 46: 471–510.
- Jones KN, Reithel JS (2001) Pollination-mediated selection on a flower color polymorphism in experimental populations of *Antirrhinum* (Scrophulariaceae). Am J Bot 88: 447–454.
- Beardsley PM, Yen A, Olmstead RH (2003) AFLP phylogeny of Mimulus section Erythranthe and the evolution of hummingbird pollination. Evolution 57: 1397-1410.
- Chittka L, Thomson JD, Waser NM (1999) Flower constancy, insect psychology, and plant evolution. Naturwissenschaften 86: 361–377.
- Gumbert A, Kunze J, Chittka L (1999) Floral colour diversity in plant communities, bee colour space and a null model. Proc R Soc Lond B 266: 1711–1716.
- Tan KK, Nishida R, Toong YC (2002) Floral synomone of a wild orchid, Bulbophyllum cheiri, lures Bactrocera fruit flies for pollination. J Chem Ecol 28: 1161–1172.
- 24. Schiestl FP, Ayasse M (2002) Do changes in floral odor cause speciation in sexually deceptive orchids? Plant Syst Evol 234: 111–119.
- Dufay M, Hossaert-McKey M, Anslett MC (2003) When leaves act like flowers: How dwarf palms attract their pollinators. Ecol Lett 6: 28–34.
- Herrera CM (1993) Selection on floral morphology and environmental determinant of fecundity in a hawkmoth pollinated violet. Ecol Monogr 63: 251–275.
- 27. Alexandersson R, Johnson SD (2002) Pollinator-mediated selection on flower-tube length in a hawkmoth-pollinated gladiolus (Iridaceae). Proc R Soc Lond B 269: 631–636.

- Haber WA, Frankies GW (1989) A tropical hawkmoth community: Costa Rican dry forest Sphingidae. Biotropica 21: 155–172.
- Pleasants JM, Waser NM (1985) Bumblebee foraging at a "hummingbird" flower: Reward economics and floral choice. Am Midl Nat 114: 283–291.
- Suzuki K, Dohzono I, Hiei K, Fukuda Y (2002) Pollination effectiveness of three bumblebee species on flowers of *Hosta siegoldiana* (Liliaceae) and its relation to floral structure and pollinator sizes. Plant Spec Biol 17: 139–146.
- Fetscher AE, Rupert SM, Kohn JR (2002) Hummingbirds foraging position is altered by the touch sensitive stigma of bush monkeyflower. Oecologia 133: 551–558.
- Lasso E, Naranjo ME (2003) Effects of pollinators and nectar robbers on nectar production and pollen deposition in *Hamelia patens* (Rubiaceae). Biotropica 35: 57-66.
- Stang M, Klinkhamer PGL, van der Meijden E (2006) Size constrains and flower abundance determine the number of interactions in a plant-flower visitor web. Oikos 112: 111–121.
- 34. Kato M (2000) Anthophilous insect community and plant-pollinator interactions on Amami Islands in the Ryukyu Archipelago, Japan. Contrib Biol Lab Kyoto Univ 29: 157–252.
- Yamazaki K, Kato M (2003) Flowering phenology and antophilous insect community in a grassland ecosystem at Mt. Yufu, Western Japan. Contrib Biol Lab Kyoto Univ 29: 255–318.
- Dupont YL, Hansen DM, Olesen JM (2003) Structure of a plant-flowervisitor network in the high-altitude sub-alpine desert of Tenerife, Canary islands. Ecography 26: 301–310.
- Vázquez DP, Stevens RD (2004) The latitudinal gradient in niche breath: Concepts and evidence. Am Nat 164: E1–E19.
- 38. Vázquez DP, Aizen MA (2006) Community-wide patterns of specialization in plant-pollinator interactions revealed by null models. In: Waser NM, Ollerton J, editors. Plant-pollinator interactions: From specialization to generalization. London: The University of Chicago Press. pp. 200–219.
- Rodríguez-Gironés MA, Santamaria L (2006) A new algorithm to calculate the nestedness temperature of presence-absence matrices. J Biogeogr 33: 924–935.
- 40. McGill B (2003) Strong and weak tests of macroecological theory. Oikos 102: 679–685.
- 41. Hubbell SP (2001) The unified neutral theory of biodiversity and biogeography. Princeton: Princeton University Press. 375 p.
- Ollerton J, Johnson SD, Cranmer L, Kellie S (2003) The pollination ecology of an assemblage of grassland asclepiads in South Africa. Ann Bot 92: 807–834.
- 43. Stebbins GL (1974) Flowering plants. Cambridge: Belknap Press. 327 p.
- 44. Müller H (1883) The fertilisation of flowers. London: Macmillan. 669 p.
- Waser NM, Fugate ML (1986) Pollen precedence and stigma closure: A mechanism of competition between *Delphinium nelsonii* and *Ipomopsis* aggregata. Oecologia 70: 573-577.
- 46. Galen C (1999) Flowers and enemies: Predation by nectar-feeding ants in relation to variation in floral form of an alpine wild flower, *Polemonium* viscosum. Oikos 85: 426-434.
- 47. Brown K (2002) A compromise on floral traits. Science 298: 45-46.
- Galen C, Butchard B (2003) Ants in your plants: Effects of nectar thieves on pollen fertility and seed siring capacity in the alpine wild flower, *Polemonium viscosum*. Oikos 101: 521–528.
- Irwin RE, Adler LS, Brody AK (2004) The dual role of floral traits: Pollinator attraction and plant defense. Ecology 85: 1503–1511.
- Rodríguez-Gironés MA, Santamaria L (2005) Resource partitioning among flower visitors and evolution of nectar concealment in multi-species communities. Proc R Soc Lond B 272: 187–192.
- Ollerton J, McCollin D, Fautin DG, Allen GR (2007) Finding NEMO– Nestedness Engendered by Mutualistic Organisation in anemonefish and their hosts. Proc R Soc Lond B. In press. doi:10.1098/rspb.2006.3758
- Herrera CM (2002) Seed dispersal by vertebrates. In: Herrera CM, Pellmyr O, editors. Plant-animal interactions. An evolutionary approach. Oxford: Blackwell. pp. 185–208.
- 53. Press WH, Teukolsky SA, Vetterling WT, Flannery BP (1992) Numerical recipe in C. The art of scientific computing. Cambridge: Cambridge University Press. 1,020 p.
- Atmar W, Patterson BD (1993) The measure of order and disorder in the distribution of species in fragmented habitat. Oecologia 96: 373–382.
- 55. Atmar W, Patterson BD (1995) The nestedness temperature calculator: A visual basic program, including 294 presence-absence matrices [computer program]. University Park (New Mexico): AICS Research. Available: http://aics-research.com/nestedness/tempcalc.html. Accessed 27 April 2006.